Corporate governance in practice: the role of practitioners' understanding in implementing compliance programs

Compliance program practices

887

Received 12 August 2016 Revised 25 August 2017 16 October 2019 31 January 2020 Accepted 3 February 2020

Riccardo Stacchezzini, Francesca Rossignoli and Silvano Corbella Department of Business Administration, University of Verona, Verona, Italy

Abstract

Purpose – This article investigates the implementation of a compliance programme (CP) in terms of how practitioners conceive of and execute the responsibilities arising from this corporate governance mechanism. **Design/methodology/approach** – This study involves a practice lens approach forms the case study analysis and interpretation, involving both interviews and documentary materials collected from an Italian company with prolonged compliance experience. Schatzki's (2002, 2010) practice organisation framework guides the interpretation of CP as a practice organised by rules, practical and general understandings and teleoaffective structures.

Findings – CP practice evolves over time. A practical understanding of daily actions required to accomplish the CP and a general understanding of the responsibilities connected with the CP, such as the attitudes with which the CP is performed, are mutually constitutive and jointly favour this evolution. Dedicated artefacts – such as IT platforms, training seminars and compliance performance indicators – help spread both of these types of understanding. These artefacts also align practitioners' general understanding with the CP's teleoaffective structures imposed, including the CP's assigned objectives and the desired reactions to them. **Research limitations/implications** – The findings have theoretical and practical implications by revealing the relevance of practitioners' understanding of corporate governance mechanisms in their implementation

Originality/value – This study reveals the potential benefits of practice lens approaches in corporate governance studies. It responds to the call for qualitative studies that demonstrate corporate governance as implemented in daily activities.

Keywords Compliance programs, Corporate governance, Practice implementation, Practical understanding, Practice theory

Paper type Research paper

1. Introduction

Corporate governance (CG) is broadly defined as the "procedures and processes according to which an organisation is directed and controlled, [coherently with] the distribution of rights and responsibilities among the different participants in the organisation" (Organisation for Economic Co-operation and Development—OECD, 2007, p. 151). Furthermore, CG has been

© Riccardo Stacchezzini, Francesca Rossignoli and Silvano Corbella. Published by Emerald Publishing Limited. This article is published under the Creative Commons Attribution (CC BY 4.0) licence. Anyone may reproduce, distribute, translate and create derivative works of this article (for both commercial and non-commercial purposes), subject to full attribution to the original publication and authors. The full terms of this licence may be seen at http://creativecommons.org/licences/by/4.0/legalcode

The authors thank prof. Lee Parker for the support during the review process, and the reviewers for their insightful comments. The authors acknowledge the availability of the interviewees at "Pharma 101", and particularly thank the compliance officer and CEO in charge at the time of the interviews. The authors benefitted from comments provided by Alessandro Lai and participants at the APIRA Conference (Melbourne, Australia, 13-15 July 2016), the EIASM Workshop on Corporate Governance (Milan, Italy, 27-28 October 2016), and the research seminar held by Riccardo Stacchezzini at University of East Anglia (Norwich, UK, 13 February 2019).



Accounting, Auditing & Accountability Journal Vol. 33 No. 4, 2020 pp. 887-911 Emerald Publishing Limited 0951-3574 DOI 10.1108/AAAJ-08-2016-2685 increasingly scrutinised by critical and interpretive accounting scholars (Ahrens et al., 2011; Gendron, 2018; Parker, 2018), also in historical perspective (Shah and Napier, 2019). They have aimed to broaden the CG research scope by adopting theoretical frameworks that might surpass a schematic interpretation of CG functions (Roberts et al., 2005; Parker and Nielsen, 2011). Similarly, there are calls for qualitative research that can open the "black box" of CG (Ahrens and Khalifa, 2013) and explore how practitioners conceive of and execute the responsibilities related to CG roles and mechanisms (Parker, 2018). Indeed, "[t]he processes by which perceptions are conditioned, attitudes formed, behaviours practised and power and control are wielded (often being the prime interest of journalists, regulators and the wider community), are seemingly ignored by accounting researchers claiming to advance the cause of effective corporate governance" (Parker, 2018, p. 284). Furthermore, accounting scholars should aim to enrich the "traditional" views of CG, which are predominantly based on an agency theory approach and almost exclusively focus on resolving conflicts of interest between corporate management and stakeholders (Brennan and Solomon, 2008). Even historical perspectives on CG demonstrate a great potential, beyond the accounting history research agenda. Indeed, the variety of settings and the modalities of CG practices observable in the past can illuminate our understanding of contemporary CG and its relations with accounting (Lai et al., 2019, p. 328).

Some studies have seized the opportunity to embrace broader research approaches to CG beyond those that are traditional, predictive and predominantly quantitative (Brennan and Solomon, 2008; Hambrick et al., 2008; McNulty et al., 2013; Parker, 2018) by elucidating the potential of a practice lens approach in exploring how CG is implemented (Ahrens et al., 2010, 2011; Ahrens and Khalifa, 2013; Brennan and Kirwan, 2015). A "practice turn" appears in various social science disciplines (Cetina et al., 2001; Whittington, 2006, 2011; Simpson, 2009; Brown, 2012; Schatzki, 2018); furthermore, practice theory is also helpful in raising many context-specific research questions regarding CG (Ahrens et al., 2011, p. 320). Moreover, interpreting CG as a practice opens the "black box" of organisations by providing an understanding of corporate boardrooms' actual functions, among other governance structures and mechanisms (Ahrens and Khalifa, 2013). This also facilitates a focus on how practitioners perceive, interpret and adapt CG rules to convert them into activities (Brennan and Kirwan, 2015, p. 469). This may also enhance our understanding of how practitioners at different levels – such as boards of directors, management teams and other organisational members – respond to the regimes of accountability imposed by CG rules, and in practicing these rules, they ultimately change them (Ahrens et al., 2010, p. 10).

The practice lens approach developed by Theodore R. Schatzki (1996, 2001, 2002, 2005, 2006, 2010), an eminent professor of philosophy, has been particularly helpful in exploring CG functioning. Specifically, this has compelled scholars to capture the collective structures of knowing, reasoning and understanding (Reckwitz, 2002) that transcend single individuals and pertain to CG practices. Scholars interpreting CG practices as sets of actions practitioners "mentally organise" (Schatzki, 1996, p. 56), can elucidate how these actions are "organised" by practitioners' understanding of CG objectives, rules and duties. Ahrens et al. (2011, p. 231) particularly urge an exploration of practitioners' "practical understanding" regarding how to execute CG-related interactions, processes and tasks; without such an understanding, "corporate governance remains an incomplete practice". Nevertheless, the exhortations to adopt practice theory to empirically explore CG practices (Ahrens et al., 2010, 2011; Ahrens and Khalifa, 2013; Brennan and Kirwan, 2015) have yet to be heard. Although there is a growing number of accounting scholars that have adopted Schatzki's theory to provide empirical insights on accounting and risk management practices (Ahrens and Chapman, 2007; Jørgensen and Messner, 2010; Nama and Lowe, 2014; Bui et al., 2019), to the best of our knowledge, no prior studies has incorporated Schatzki's approach using CG practices as an empirical setting. This may be due to difficulties in collecting insights about the actual functioning of CG processes (Parker, 2018). The present research aims to fill this gap.

This research's empirical setting involves the corporate implementation of a compliance programme (CP) or a CG mechanism that requires all company employees, managers and directors to be accountable for certain actions and controls according to the behavioural rules established by an organisation's overall codes of conduct (Weaver et al., 1999b). The present article draws on Schatzki's practice theory to develop a theoretically informed narrative of the empirical materials provided by an Italian pharmaceutical company, henceforth "Pharma 101", for confidentiality. This company adopted a CP starting in 2005 in accordance with the Legislative Decree No. 231 of 8 June 2001, an Italian regulation introducing corporate crime liability [1], hereafter "LD 231". Applying a CP has considerable relevance in this context for several reasons that relate to its location as this Italian company is subject to the corporate liability penalties introduced by LD 231 (i.e. financial sanctions and debarment from business activities); industry, as the Italian pharmaceutical industry is highly regulated, and the Italy's Association of Pharmaceutical Companies provides CP guidelines that Pharma 101 follows; and ownership structure, as Pharma 101 is controlled by a global healthcare company that requires its subsidiaries to adhere to its ethical and compliance policies.

This analysis parallels Schatzki's work by exploring how CP implementation unfolds as a practice "organised mentally" by practitioners (Schatzki, 1996, p. 56). In particular, the research elucidates how practitioners conceive of and execute CP-related responsibilities. As the CP is a CG mechanism that involves engagement by all corporate employees, managers and directors while they execute their activities, we posit that investigating CP implementation processes through a practice lens approach particularly contributes to elucidating CG's actual daily functioning (Ahrens *et al.*, 2011; Ahrens and Khalifa, 2013; Roberts *et al.*, 2005; Parker, 2018). Intrinsically, the study has both theoretical and practical implications by offering empirical insights on practitioners' understanding of CG mechanisms in their implementation process.

This article proceeds as follows: Section 2 introduces CP logic and reviews extant literature on CP practices. Section 3 presents this study's theoretical framework and Section 4 describes the methodology, including the empirical setting and how the evidence was collected and analysed. Next, Section 5 contains the main findings from the case study of CP implementation. Section 6 discusses the main findings, and Section 7 concludes by highlighting this study's contributions, some limitations and suggestions for further research.

2. Compliance programme practices

Compliance programmes have become increasingly prevalent worldwide and especially following the passage of various national regulations – such as the United States' Sarbanes-Oxley Act, Australia's Corporate Law Economic Reform Program Act and South Africa's King Report on Corporate Governance – in response to fraud scandals (Francis and Armstrong, 2003; Beale and Safwat, 2004; Svensson et al., 2009). These types of CG mechanisms have been particularly emphasized in such regulated trades as the financial, insurance and pharmaceutical industries (Bartrum and Bryant, 1997). Furthermore, several international agreements have been established in response to growing corruption due to markets' opening and globalisation processes including the following: the Convention Protecting the European Union's Financial Interests – Fight Against Fraud (Brussels, 26 July 1995) and its first protocol (Dublin, 27 September 1996); the Convention Against Corruption Involving Officials (Brussels, 26 May 1997) and the OECD's Convention on Combating Bribery of Foreign Public Officials in International Business Transactions (Paris, 17 December 1997). In line with these agreements, some countries have adopted corporate criminal liability regulations, such as the Italian LD 231; Spain's 2003 Ley Orgànica Numero 15, with updates in 2010 and in 2015 and the United Kingdom's Bribery Act of 2011. Each national regulation influences CP in coordination with the local penal system. Therefore, the facilities that companies may use to implement a CP meet locally regulated legal requirements.

Moreover, CPs create regimes of accountability (Volkov, 2015; Jones, 2017) under which all employees are required to standardise their behaviour "within the domains of ethics and legal compliance" (Weaver et al., 1999a, p. 42). The CP also assigns them various roles and related responsibilities to prevent misconduct. As individuals or body members, the employees involved include the board of directors and CEO; the "supervisory body" to which LD 231 assigns the monitoring of CP implementation; members of departments, such as the compliance officer, internal auditing and legal officer and managers and employees working in business departments [2]. Within their oversight functions, governance bodies are also required to train employees and managers in codes of conduct and compliance with these codes; exert (third-level) controls to avoid misconduct and establish disciplinary systems to be applied in the case of misconduct. Specifically, the board of directors is responsible for adopting the CP, and the CEO is directly involved in the necessary actions towards its implementation. The supervisory body monitors CP implementation and reports its effectiveness to the board of directors. Staff departments can support the supervisory board and CEO by performing controls in their daily activities. Managers of business departments are responsible for their activities as well as those performed by the employees within their departments. Consequently, business managers impose (second-level) controls on employees' activities. However, employees are personally responsible for their individual actions and should implement concurrent (first-level or self-) controls that deter misconduct.

Academics have explored CP practices from various research perspectives, including business ethics and business law and management (e.g. Weaver and Treviño, 1999; Schwartz, 2004; Arjoon, 2005; Parker, 2006). However, such practices remain unexplored within accounting literature (Brennan and Solomon, 2008). Previous studies highlight the need to investigate practitioners' motivations, perceptions and incentives to understand how these programs function. For example, Weaver and Treviño (1999) demonstrate that employees' perceptions of programs affect how these programs are implemented. When employees perceive CP as a program to help their decision-making, with ethical advice and counselling, they are more prone to engage in its implementation and report misconduct. MacLean and Behnam (2010) highlight the dangers of decoupling CP from core business activities, as practitioners respond to this decoupling in their CP implementation. According to Kaptein (2011a, b), codes of misconduct are embedded among senior managers and other employees; furthermore, Schwartz (2004) demonstrates that codes of conduct's content and procedural aspects – such as their examples, senior management support and training – influence practitioners' perceptions and behaviours. Similarly, Singh (2011) indicates that managerial perceptions of these codes' "effectiveness" in deterring misconduct depend on how the codes are implemented and communicated within organisations.

In summary, these studies highlight that practitioners' motivations, perceptions and incentives affect CP implementation, and how CPs are implemented affects their behaviour. However, no prior research to the best of the authors' knowledge has explored how practitioners perceive the CP during its execution. In Schatzki's (2001) words, no prior studies have detailed how CP implementation unfolds as a practice that practitioners mentally organise. Thus, we draw on practice theory to fill this gap as we are unaware of any prior studies that have adopted a practice lens approach to investigate this CG mechanism.

3. Theoretical frame

This section introduces the key concepts Schatzki developed in his approach to practice theory. It also reveals pertinent key studies adopting Schatzki's works as a theoretical reference and highlights practice theory's potential in CG research.

3.1 Schatzki's approach to practice

Since the 1980s, contemporary social theory has taken a "practice turn", largely due to seminal works by such distinguished scholars as Pierre Bourdieu, Michel de Certeau, Michel

Foucault and Anthony Giddens (Whittington, 2006; Miettinen *et al.*, 2009; Brown, 2012; Hui *et al.*, 2017; Schatzki, 2018). As common denominators, these theorists examine the consequentiality of everyday actions in the production of the social order, the rejection of dualism and the dualities of structure and agency and cognition and action and mind and body (Feldman and Orlikowski, 2011). Moreover, practice theory posits that "social orders (structures, institutions, routines, etc.) cannot be conceived without understanding the relevance of agency in producing them, and similarly, agency cannot be understood 'simply' as human action, but rather must be understood as always already configured by structural conditions" (Fieldman and Orlikowski, 2011, p. 1242).

Schatzki is a "central interlocutor" of practice theory (Caldwell, 2012, p. 2), as his approach to practice is rooted in the idea that "social order is established within the sway of social practices [...] and mind is a central dimension of this "process" (Schatzki, 2001, p. 50). This is because the "mind is a "medium" through which practices are organised" (Schatzki, 2001, p. 61), Specifically, Schatzki's (2002, 2010) "practice organisation framework" describes practices as sets of actions linked through and organised by four principles, also labelled "properties" or "dimensions" as follows: practical or general understandings, rules and teleoaffective structures [3]. First, practical understanding involves the abilities related to actions that comprise a practice, which implies that one will know how to act regarding some given practice. Individuals equipped with these abilities know what to do to perform a certain practice or "knowing how to X, knowing how to identify X-ings, and knowing how to prompt as well as respond to X-ings" (Schatzki, 2002, p. 77, emphasis in the original). Furthermore, Schatzki (2002, p. 88) notes that practical understandings alone rarely determine action, except for "disperse practices" or such simple actions typically organised only through practical understanding as acts of describing, ordering, explaining and reporting. "Integrative practices" involve "multiple actions, projects, and emotions" (Schatzki, 2002, p. 88) and typically also require other organising properties.

Second, rules are explicit formulations that combine particular actions, such as laws, rules of thumb or unwritten rules and other prescriptive judicial remedies. They are formulated and applied to either control current activities or cause specific effects. Rules connect the action of a practice to the extent that actors adhere to it: "rules do not determine what people do; rather, what people do determines what following rules amounts to" (Schatzki, 1996, p. 298). However, similar to practical understanding, rules "only intermittently and never *simpliciter* determine what people specifically do" (Schatzki, 2001, p. 60).

Third, teleoaffective structures include two components including teleology and affectivity. Teleology refers to the goals in performing a practice, while affectivity involves its accepted or prescribed emotions and moods. In summary, a teleoaffective "structure' is a range of normativised and hierarchically ordered ends, projects and tasks, to varying degrees allied with normativised emotions and even moods" (Schatzki, 2002, p. 80). Even if evenly incorporated into actors' minds and actions, teleology and affectivity are the practice's properties rather than those of the actors. These also refer to the objectives and emotions that are deemed mandatory or acceptable. Fourth, general understandings represent an overall holistic understanding, and are expressed as the manner in which people carry out projects and tasks (Schatzki, 2002, p. 86).

Variations in the practice organisation's principles determine how their practices evolve, as they also evolve with time in response to practice-related events (Schatzki, 2002, p. 104): "practices are not static. They evolve as circumstances change, opportunities and problems arise, personnel changes, new ideas arise, and so on" (Schatzki, 2005, p. 475). The actions that constitute practices are inescapably – and often fundamentally – entwined with "material entities" (Schatzki, 2002, p. 16), or *people* (human beings), artefacts ("products of human action"), organisms ("life forms other than humans") and things ("non-living entities whose being is not the result of human activity") (Schatzki, 2002, p. 22). These material entities facilitate such practices to "transpire" and work with them to mould the social order (Schatzki, 2005).

3.2 The critique, adoption and potential of Schatzki's approach

Schatzki's approach to practice has been subject to various critiques (Caldwell, 2012; Welch and Warde, 2017) that relate to both its ontological and epistemological stances. For instance, Caldwell (2012) argues that Schatzki's account not only overlooks the relationship between agency and practice but also underestimates the relevance of discourse and language in moulding practices. Furthermore, critiques refer to the opacity of some concepts and particularly that of general understanding (Caldwell, 2012; Welch and Warde, 2017).

With these critiques, several scholars have begun to operationalise and refine Schatzki's account. Specifically, Schatzki's approach to practice has made its entrance into the accounting literature due to Ahrens and Chapman's (2007) pioneering contribution. Jørgensen and Messner (2010) later indicated that accounting can help mould practitioners' general understanding within the strategic context of new product development practices. Furthermore, Nama and Lowe (2014) contributed to the understanding of accounting within private equity practices by theoretically advancing the concept of general understanding. More recently, Bui *et al.* (2019) demonstrated how general understanding has forged changes in risk management practices.

Some scholars have also elucidated the practice lens approach's potential in studying CG (Ahrens *et al.*, 2010, 2011; Ahrens and Khalifa, 2013; Brennan and Kirwan, 2015). Moreover, the practice lens has been deemed helpful in "captur[ing] the dynamic, interactive nature of governance", which consequently limits the "dangers" of studying its "appearance" (Brennan and Kirwan, 2015, p. 467). From a practice theory perspective, such "structural" elements of CG as its regulations, firm governance policies and daily policies and procedures (Brennan and Kirwan, 2015, p. 470), are interpreted as "resources of a practice that practitioners, in drawing upon them, can change [. . .] through the adoption and adaptation of activities, ideas and tools" (Ahrens *et al.*, 2011, p. 318). Practitioners are not passive subjects of CG implementation processes as they "respond to regimes of accountability and can contribute [to] shaping them" (Ahrens *et al.*, 2010, p. 10).

Moreover, studying how CG operates "in practice" involves investigating the practitioners' capabilities and initiatives to translate governance rules into organisational activities (Whittington, 2006; Brennan and Kirwan, 2015) according to how they conceive of governance roles and actually execute their related responsibilities (Parker, 2018). However, rather than focusing on individuals' perceptions, motivations and incentives, research influenced by practice theory has adopted CG practices as a "unit of analysis" (Ahrens *et al.*, 2010, p. 2) to demonstrate how CG practitioners share an understanding of the actions to execute CG mechanisms. This should provide a more thorough perspective of CG functioning given that "if [corporate governance] wants to become a functioning practice, [it] must develop practical understandings that enable practitioners to connect principles and rules more effectively. Practical understandings are complexes of know-how, and they have been neglected in the corporate governance discussion thus far" (Ahrens *et al.*, 2011, p. 319). Schatzki's (2002) account of practice should facilitate analyses of the practical understanding of CG mechanisms as this account perceives practical understanding as a practice-organising principle.

Despite practice theory's highlighted potential for CG research, no prior studies to the best of our knowledge are based on such direct empirical explorations. Indeed, Brennan and Kirwan (2015) reviewed prior studies on audit committees, while Ahrens *et al.* (2010) developed their arguments based on prior research of boards of directors and audit committees. The contribution by Ahrens *et al.* (2011) is deductive and not empirically based, while Ahrens and Khalifa (2013) underline the importance of studying CG practices based on a review of three qualitative field studies. This study fills this gap by adopting Schatzki's practice lens approach to empirically explore how the CP practice unfolds as sets of activities that practitioners "mentally organise" (Schatzki, 1996, p. 56).

Compliance

4. Methodology

This section describes the empirical setting by introducing the case study company, its CP practices and the regulatory context in which the company applies this CP. The section then explains how evidence was collected and analysed to develop a theoretically informed case narrative.

4.1 The embirical setting

This study explores CP practices by examining empirical materials provided by Pharma 101, an Italian pharmaceutical company that adopted a CP in accordance with LD 231 in 2005; the company routinely updates its CP to match legislative updates and the organisation's specific needs. Pharma 101 belongs to one of the largest pharmaceutical groups worldwide and delivers health solutions through its prescription medicines, vaccines, biological therapies and animal health products, which it markets both directly and through its joint ventures. Currently, Pharma 101 directly employs approximately 200 people and indirectly employs an additional 1,200 people through its Italian subsidiaries.

Applying a CP is considerably relevant to Pharma 101 for several reasons. First, as an Italian company, Pharma 101 is subject to the corporate liability introduced by LD 231. Specifically, LD 231 offers companies a means to avoid responsibility in the case of employee misconduct, if they can prove it despite what any individual employee has done; the company has implemented a type of CP specified by LD 231 or the purported "231 model" [4, 5]. Considering the benefits from adopting this CP, and although compliance with LD 231 through CP is not compulsory, this practice is widespread among Italian-listed and other large companies (Allegrini and D'Onza, 2003; Arena and Azzone, 2007; Corbella, 2013; Rossignoli, 2013; PwC, 2017). When twice charged with corporate crimes according to LD 231, Pharma 101 successfully demonstrated that the employees' conduct was inspired not by the company but by the private interests of the individuals in question. Therefore, its CP operated as a defence against corporate liability penalties.

Second, Pharma 101 operates in a regulated pharmaceutical industry, with peculiarities in terms of the nature of the products, the stakeholders involved and the multiple economic and social interests associated with its products (Dukes, 2002). The bribery of public officers (doctors) is one of the most representative examples of corporate crime conducted by pharmaceutical companies. Specifically, pharmaceutical representatives could potentially bribe healthcare professionals during their informative duties aimed to promote the company's drugs and compel them to prescribe the company's drug to patients instead of an alternative. In the Italian pharmaceutical industry, Farmindustria – Italy's association of pharmaceutical companies and a member of the Italian industrial federation – prepares and updates detailed guidelines that illustrate what a CP should consist of in considering the industry's peculiarity. This is characterised by the unique nature of its products, participants and varied economic and social interests. Pharma 101 has followed these guidelines in designing and implementing its CP.

Third, Pharma 101's CP adherence was bolstered by the ethical stance of its parent company; its mission statement proclaims an utmost respect for humans, health and ethics. This parent company is a global-listed healthcare firm that operates in several countries and controls one of the world's largest pharmaceutical groups. Pharma 101 must adhere to its parent company's ethical and compliance policies.

4.2 Data collection

To achieve an understanding of CP implementation as a practice, we based our analysis on interviews with individuals who are not only engaged in the CP's daily functioning within the

company at different organisational levels but also belong to different departments, for example, such business departments (as sales, marketing, meeting services) and such staff departments (as compliance and legal ones).

The interview process spanned two years and included 21 interviews. Similar to prior research investigating the evolution of practices with a practice lens approach (Shove and Southerton, 2000; Shove and Pantzar, 2005), the interviews occurred after the practices had already begun. The interview process was developed in two phases as follows: the first phase began in March 2015 and finished in September 2015 and comprised 16 interviews. The second phase began in November 2016 and ended in February 2017, and it comprised five interviews [6].

Each interview was recorded with permission and lasted 45 min on average. At least two of the three authors participated in each interview by submitting questions and taking notes, which were used in addition to the interview transcripts. Each interview began with an enquiry related to the interviewee's background; the researchers also explained the project's general objective (i.e. to draft a paper with empirical evidence of processes related to CP implementation as associated with LD 231/01). Each interviewee was individually contacted and interviewed after obtaining his/her office phone number and business e-mail from Pharma 101's organisational directory, which is also a fundamental element of the CP. Interviewees' anonymity was guaranteed as their identities remained classified and none of their information was shared with anyone else in the company. These individual interviews also deterred any reciprocal influences among respondents.

The first interviews included the compliance officer as well as members of the supervisory body with primary competencies centred on policies and procedures, business compliance, risk management and internal auditing. These interviews provided the authors with some familiarity as to the governance structure. With this "formal" CP functioning and governance structures in mind, the authors conducted further interviews. As the analysis focused on the practical implementation of CP throughout the company, several interviews included managers and employees working for the company's business departments. In their capacity as a business division's chiefs or employees, they must accomplish core business tasks while avoiding misconduct and engaging in the (self-) controls established in the CP. The staff department interviews included members of the compliance and legal offices; two supervisory body members also provided insights. Across this broad sampling, these interviews included employees who experienced all phases of the CP implementation, as well as some who began their employment with Pharma 101 more recently; we interviewed employees with a minimum five years' seniority at the time of the interview. None of the interviews included employees who worked for the company previously but left before the data collection period although such a sample would only include six people who resigned after the CP's implementation. Table 1 summarises the interviewees' details, and Table 2 outlines the main issues covered during the interviews. The identifying code of the interviewees in Table 1 is based on the department's area (staff or business) and their position (manager, employee, etc.).

4.3 Data analysis

An interpretative stance guided by Schatzki's practice theory was adopted in analysing the collected data and developing a theoretically informed narrative (Auerbach and Silverstein, 2003). Codes were generated according to the theoretical framework and discussed among the authors (Strauss and Corbin, 1990; Auerbach and Silverstein, 2003). Specifically, CP implementation has been investigated in terms of the daily activities enacted by practitioners and the practice-organising principles Schatzki (2002) depicted the following: (1) the knowledge of how to act relative to the CP (practical understanding); (2) the rules activated to

Role	Department	Department's area	Position	Background	Number of interviews	Coding	Compliance program
Compliance officer	Compliance office	Staff	Manager	Finance and audit	4	SM1	practices
Legal officer	Legal office	Staff	Manager	Legal	1	SM2	
Compliance officer's assistant	Compliance office	Staff	Employee	Compliance and legal	3	SE1	895
Compliance employee	Compliance office	Staff	Employee	Legal	1	SE2	
External member of the supervisory body	Not applicable	Staff	External expert	Administration	2	SEE1	
External member of the supervisory body	Not applicable	Staff	External expert	Administration	1	SEE2	
CEO	Not applicable	Business	Executive	Administration	1	CEO	
Sales manager	Sales	Business	Manager	Marketing and sales	1	BM1	
Conferences manager	Meeting services	Business	Manager	Marketing and sales	1	BM2	
Marketing manager of the oncology business unit	Oncology	Business	Manager	Medicine	1	BM3	
Medical manager's assistant	Marketing	Business	Employee	Medicine	1	BE1	
Business director's assistant	Marketing	Business	Employee	Administration	1	BE2	
Pharmaceutical representative	Regional sales	Business	Employee	Medicine	1	BE3	
Pharmaceutical representative	Regional sales	Business	Employee	Medicine	1	BE4	
Pharmaceutical representative	Regional sales	Business	Employee	Medicine	1	BE5	Table 1. Interviewees

Knowledge of the company's CP and the law that led to the CP's introduction

Knowledge and perceptions of the company's efforts to implement and adapt the CP over time

Offices/organisational members primarily involved in the CP's implementation

Rules and procedures introduced by the CP

Personal duties arising from the CP

Understanding of the responsibility derived from the CP's implementation

Understanding of the activities pertaining to the CP's implementation

Interactions and discussions with colleagues as a consequence of the CP

Personal perceptions and feelings about the CP and its requirements

Opinions of other organisational members' perceptions and feelings about the CP and its requirements Awareness of compliance-related lawsuits against the company and their impacts on employees' behaviour Artefacts (e.g. training activities, internal reports and reviews and e-mails) mobilised as a consequence of the CP

Table 2. Key issues discussed during the interviews

regulate the CP's adoption and to which practitioners adhere, such as laws regarding corporate crimes and "Model 231", the industry's CP guidelines and corporate codes of conduct; (3) the intentions, beliefs, desires, hopes and expectations associated with the CP (teleoaffective structures); (4) the attitude with which the CP is implemented (general understanding). Other codes relate to the people involved in implementing the CP and to the artefacts that materialise during the CP's implementation process, such as internal controls and risk management systems, training courses, internal reports and reviews of internal controls, the CP reports produced by consultancy firms and other formal or informal communications among the various actors involved in CP implementation. Additional codes were also generated during the analysis to facilitate a better understanding of practice variations, such as time and change.

All the interviews were simultaneously transcribed and analysed by two authors, with a careful focus on how the actors described and interpreted their activities to execute the CP (Orlikowski, 2002). To detect possible divergent perceptions of employees hired at different times, the authors carefully reviewed the interview transcripts in light of the hiring time. Such review allowed confirming that there were no notable dissimilarities in the personal perceptions of employees who were hired at different times, rather, their perceptions of the investigated topics, as specifically derived in this study, were consistent regardless of their duration with the company.

Specifically, interviewees shared the same views about the CP implementation phases. These three phases, that are presented separately in three sub-sections within the following Section 5, were first identified according to the interviewees' perceptions of their activities. The analysis of the interviews informed by the Schatzki's practice theory allowed us to confirm this timeframe because we noticed significant differences in how the practice properties manifested across these phases.

This study complements the interview analysis by relying on documents related to the CP, whether public or private. These documents supported efforts to verify the interviewees' memories about old facts and represented various artefacts related to the CP implementation process, including: lists of offences, lists of compliance risks, crime prevention procedures, disciplinary penalties (withholding on salary, formal reprimands, demotions, dismissal), the supervisory body's composition and performance, the corporate processes involved in internal control systems for LD 231, training programs, program compliance assessments and the functional management of information flows.

5. Findings

This section provides a theoretically informed narrative of the implementation of the LD 231–related CP within the case study company. It particularly highlights how CP implementation has evolved since the time of its introduction until the more recent interviews relative to changes in the practice-organising principles Schatzki (2002, 2010) observed: practical and general understandings, rules and the teleoaffective structure. The section is separated into three sub-sections according to the three phases of the CP implementation that emerged from this analysis.

5.1 Phase 1: the CP practice as abiding by restrictions

In July 2005, Pharma 101's board of directors decided to introduce a CP associated with LD 231 with the aim to ensure the company's compliance with the new law and emerging best practices. Although a CP was not required by law, the board of directors decided to adopt it four years after the related legislative decree was released. As the compliance officer (SM1) and the compliance officer's assistant (SE1) both noted, most of Pharma 101's competitors began to address this regulation, and Pharma 101 subsequently intended to do the same.

The CP included a code of conduct, several procedures for handling business activities, controls over business activities and penalties for violations. As required by LD 231, the board of directors appointed a supervisory body, in which Pharma 101 comprised the following: a legal officer, human resources officer, the CFO and the CEO. Initial actions involved an internal audit by finance and legal managers with the support of an external lawyer, which included: (1) associating the extant business procedures, which describe how to address business activities, with specific controls; (2) matching these controls with penalties for violations and (3) coordinating these controls with the principles in the extant code of conduct.

The board of directors later created the compliance department as a separate and autonomous department that directly reports to the former. A compliance officer with a background in finance and audit management was appointed and remains in charge. Furthermore, a new legal officer with a corporate legal background was hired. Following the board of directors' requests, the compliance department first identified the additional crimes with which the company can be charged according to LD 231. It also proposed that the board of directors adopt additional controls and augmented the existing controls with some specific monitoring mechanisms to prevent misconduct. The compliance staff formalised these *rules* by performing a self-assessment devoted to identify and describe the following: (1) the double signature mechanisms, (2) the segregation of responsibilities and (3) the suppliers' selections. Moreover, specific tests were periodically conducted to verify the compliance with these additional controls. Each test's results were discussed with the board of directors, which used such periodic information inflows to observe the alignment between rules and behaviours.

While the compliance staff periodically conducted these tests, personnel within the business departments limited their efforts to respect the new rules' explicit limitations. Specifically, people read the procedures as they were written in the CP to identify the limits on the actions they intended to execute. Subsequently, employees constrained their actions according to what was allowed in the CP's procedures. For example, an additional restriction on marketing activities limited the samples that a pharmaceutical representative could provide to doctors, as minimising this quantity could avoid their improper use. As the "proper" number had to be identified according to the spirit of the CP – which encouraged the avoiding of "freebies" or other means to "win the doctors over" – the rules also provided a maximum number of samples for each doctor. Thus, pharmaceutical representatives chose to provide exactly the maximum amount of samples indicated in the rule rather than identifying the proper number of samples for each doctor.

The business department managers' and employees' practical understanding of the CP practice was limited to respect what was explicitly indicated in the individual rule. The CP practice from their perspective consisted of the following: (1) carefully reading the CP procedures to identify the actions to perform or avoid and (2) constraining their daily activities to what was clearly established in the CP procedures. This practical understanding manifested as a respect of CP restrictions that were explicitly stated in the rules that were subject to specific controls, instead of attempting to independently interpret these restrictions:

At that time, I did not realise the big picture. I mean, I was told about the new constraints and I only cared to understand what I should do, and be careful to respect it. In other words, my efforts were devoted to ensure I was slavishly adhering to the limits (BE4, a pharmaceutical representative).

Some employees within the company's staff departments were unaware of how to handle the CP's initial adoption as the board of directors only provided a limited effort in communicating the CP's adoption throughout the organisation. The only *artefact* in place in this phase consisted of newsletters devoted to notifying employees of the new limits and restrictions.

The board of directors then charged the compliance department with the first implementation activities, but those working within the compliance department had a shallow, limited understanding of the CP due to the novelty of the legislation and its underlying principles:

The first time I heard about the CP was during a cyclical meeting [...] compliance staff told us that the company had adopted the CP according to a certain Italian law, but at that time I had no idea of what it was, or what I had to do to deal with it (BE3, a pharmaceutical representative).

Well [...] at first we met in our meeting room, and it was kind of funny because we felt like we were groping in the dark. None of us had a clear idea...I mean...nobody had an idea of what a CP was, and especially me (SM2, a legal officer).

A *general understanding* of the confusion surrounding the overarching CP system transpired through daily activities. While the *teleoaffective structure* involves the willingness to commit to the new law and emerging best practices, neither business managers nor employees understood why the company chose to introduce the CP as they deemed Pharma 101 already compliant with the stringent procedures enforced according to the parent company's compliance system. The new CP was activated only to comply with the law and emerging best practices:

At that time I thought that we were missing the umpteenth rule (ironically)...about samples...It impacts not only the pharmaceutical representatives, but we even have a system linking the number of allowed samples with production to say how many samples they have to produce...and no more. [...] I had the feeling that we were already acting in an ethical way, so it was not clear to me what more we had to do than what we had already been doing. [...] The suppliers' selection process was already strictly regulated...instead, with the CP, we have been required to differentiate our actions between "preferred" and "non-preferred" suppliers: I usually had to check to make this distinction, because I did not realise the reasoning behind this (BM1, a sales manager).

At the beginning I was very confused; I did not know these rules existed and that I had to comply with them, so I felt burdened. It was frustrating (BE4, a pharmaceutical representative).

To summarise, the practice during the CP implementation's first phase only consisted of abiding by the restrictions explicitly established by the CP's new *rules*, such as procedures and controls. The CP practice reflected not only a *practical understanding*, limited to what was explicitly indicated in the rules but also a *general understanding* of the confusion regarding its overarching system. The *teleoaffective structure* aimed for alignment with the law, with emerging best practices only permeating the compliance staff's perspective, while the business staff did not share any particular belief in the practice. Newsletters devoted to advising on the newly introduced restrictions were the only *artefact*, and they were found to have a superficial impact on people's perspectives.

5.2 Phase 2: CP practice as executing compliance staff's advice

The second phase began around early 2009 and ended in late 2012, and it included business managers' and employees' direct involvement in updating the CP. External business consultants – charged with collaborating with the compliance staff to update the CP with the board of directors' approval – interviewed the business staff to map each business process and suggest to the company's management the various controls to add.

Daily activities evolved, in the sense that business departments began to interact among themselves and with the compliance staff to choose the actions to be executed. Specifically, dialogues increased among people across different business departments in an effort to brainstorm and share common experiences. Most of the business department interviewees

identified such interactions among offices as "habitual". All interviewees employed by Pharma 101 at that time recalled their efforts to address the CP.

Managers were constantly in contact with the compliance officer through e-mails, by telephone and in person to gather reassurance of what instructions to convey to the employees in conducting individual activities. They even consulted the compliance department regarding ordinary tasks and to confirm that they had comprehensively understood the meaning of the CP and that their activities were compliant with the CP's rules. For example, the conference manager scheduled a meeting with the compliance officer before organising a conference to verify that its preliminary concept was compliant with the CP. After sharing its basic characteristics with the compliance officer – such as the conference's purpose, speakers, participants, duration, sponsors and the option of setting a fee – the manager progressed with the activity by asking the compliance officer for reassurance about the activity's CP compliance as further details manifested during the process.

Similarly, the compliance department mandated that pharmaceutical representatives complete a specific form to report details of every meeting with doctors, such as the technical information given, the samples provided and clarifications required. This reporting aimed to provide an information inflow to control ordinary activities. However, instead of completing each report after each doctor's meeting, Pharma 101's representatives collected their doubts and waited until the end of the week to ask the compliance staff for suggestions regarding how to complete their weekly reports. Subsequently, the compliance officer's assistant (SE1) became accustomed to seeing business department employees queuing at their doors. Only after obtaining validation regarding such compliance could they conduct their activity. If the compliance department did not validate their proposed method, the employees instead requested specific instructions for the contingent situation and then executed what the compliance department indicated they should do.

The actions conducted by business managers and employees manifested in a *practical understanding*, which consisted of executing the compliance staff's instructions according to the CP. While the business staff could identify the problems associated with the CP, they continuously enquired regarding the proper actions to comply with the CP and asked for reassurance before performing activities that could potentially generate misconduct. People working in Pharma 101's business departments still did not understand the need for the usefulness of additional controls and suffered for the CP's unquestionable limitations to their daily activities. The consideration with which business department managers asked for the compliance officer's opinions before any decision also conveyed their fear of failing to identify the correct action:

I was always uncertain of what I had to do, so I continuously looked for suggestions from compliance (SM2, a conference manager).

They were afraid to be wrong, so they preferred asking to ensure they were compliant in what they were doing (SE1, the compliance officer's assistant).

Thus, the fear of penalty was the attitude under which the practice was performed (the *general understanding*). This attitude was particularly derived from the introduction or reinforcement of *rules*: (1) the "compliance correction" in the remuneration system and (2) the internal disciplinary system.

First, a "compliance correction" was introduced in the managers' and employees' remuneration system. Dedicated compliance performance indicators were introduced to rectify the performance evaluation, such that employees who perform well in terms of compliance score additional points. The employees with a minimum compliance correction score obtained a percentage of increase in their base salary, while managers and employees who performed poorly experienced withholdings from salary. Consequently, the compliance

AAAJ 33.4

900

correction in both cases impacts employees' performance evaluations, and thus, their salaries:

I realised I was not even free to treat doctors to a cup of coffee. If I failed, I would be penalised for that. So I tried to do my best to stand out in my performance evaluation (BE5, a pharmaceutical representative).

Second, the board of directors followed the compliance officer's suggestion and reinforced the internal disciplinary system by adding more specific penalties associated with CP violations. Some employees who had failed to execute CP practices received penalties according to this internal disciplinary system. The compliance officer (SM1) recalled that on average, the company would exhibit "soft" measures if a manager or employee performed poorly, such as moving him or her to a different business unit or another office. Furthermore, no one was ever dismissed for any business improprieties other than compliance violations; the disciplinary system includes severe measures for any non-compliance; for example, two business department employees were asked to step down.

The CEO and compliance officer both recognised the need for people to understand how to execute the CP and its overarching system.

I think that just telling them what the CP is according to the law and rules would never have led anywhere. They needed to build their awareness of what they have to do in practice. To this end, we usually show them examples of situations they would have to cope with by explaining what they would be asked to do according to the CP, and what, on the other hand, would be considered wrong (SM1, a compliance officer).

Moreover, the compliance department established a dedicated *artefact* as follows: an online course providing detailed information about the law inspiring the CP, how extant procedures have changed, the additional control introduced and several examples of incorrect behaviour taken from the press. The CEO mandated that every employee take this online course.

Nevertheless, employees were still uncertain in autonomously identifying how to choose to execute one action instead of another to comply with the CP. Thus, another artefact was established as follows: a comprehensive training program relying on interactive sections, practical examples and self-assessments. Training increased in quantity and frequency as several sections devoted specifically to the CP were added to those regularly organised in the company. Such training sections were delivered during periodic meetings to provide more concrete, real-life examples; these allowed personnel to visualise and realise the CP's role in preventing misconduct and protecting both the company and employees from undesired consequences.

The enacted artefacts intended to avoid misconduct and exhibit an overall concern for avoiding any wrongdoing or *teleoaffective structures*. Specifically, these structures are derived from the impulse of the new CEO who was really concerned with misconduct and supported the introduction of dedicated rules and artefacts. The compliance officer (SM1) observed that the CEO was particularly sensitive to the CP because he/she was affected by a lawsuit, and he/she had to personally cope with compliance problems in the company where he/she previously worked. While that company won its case, he/she is fully aware that its compliance made a difference. The CEO made considerable efforts to increase employees' awareness of the risks of crimes. He/she directly engaged with several communication initiatives towards employees. For instance, he/she regularly sent e-mails and alerts about the relevance of proper behaviour within the pharmaceutical sector and examples of corporate crimes taken from the media (e.g. newspapers and podcasts). Business managers and employees were overwhelmed by these "tone at the top" initiatives, and they started believing that CP was not an issue only to be addressed by the company's top management. This resulted in concerns of the CP's concrete impact on anticipated behaviours every time they had to set up even ordinary activities:

I remember I was bombarded by the new CEO's e-mails. At the beginning I was almost ignoring them, but when I saw that the alarm was so persistent, I started worrying about the CP. I then began reading them carefully, and gradually I realised that the CEO really cared about CP and there were reasons for that (BE5, a pharmaceutical representative).

In summary, the CP practice during the second CP implementation phase consisted of following the compliance staff's advice regarding the "proper" actions to be executed. Facilitated by the previously mentioned *artefacts*—such as the web course, training program and CEO's communication initiatives—a *practical understanding* emerged in terms of executing the compliance staff's instructions, while the *general understanding* involved the fear of penalties according to the *rules* that have were introduced or reinforced in this phase (the "compliance correction" in the remuneration and internal disciplinary systems). Those focused on avoiding incorrect actions found that additional rules did not facilitate constructive engagement with the practice. Further, this attitude toward the practice coherently developed with a *teleoaffective structure* based on concern for potential incorrect actions and a desired aim of avoiding misconduct.

5.3 Phase 3: CP practice as self-assessment of legality

The third phase began in early 2013 and continued to the interview period; the practice in this phase consisted of autonomously choosing daily activities with a concern for the legality of such activities. Most managers and employees knew what actions were appropriate without reading the procedures because they could recall them in detail. The practice includes asking for the compliance officer's opinion, but this was seldom the case for unexplored projects:

People [working in business departments] are now autonomous. They are engaged in the controls and can help us. Many people walk up to us during their coffee break to discuss specific cases, but they usually have already a clear idea about the crucial points and just want us to reassure them that they are on the right track (SE1, the compliance officer's assistant).

I am used to knocking on Compliance's door to ask for details any time I am going to engage in a new initiative, [or] any time I have new incoming doubts. Further, I informally talk to my colleagues in other departments to see what compliance was suggested for their situations and how they are behaving with reference to the CP (BM3, marketing manager of the oncology business unit).

The CP practice became completely embedded in daily activities as people respected the *rules* as fact. For example, the conference manager independently established the general conference by avoiding extravagant locations because he/she recognised that would be intended as a benefit offered to the participant doctors. This autonomy in decision-making reveals an awareness of the actions to execute. Moreover, the compliance officer (SM1) noted that they aimed to improve managers' and employees' understanding of the reasons why such controls exist and the intangible benefits derived from working for an unblemished company. The compliance officer and supervisory body members recognised that these efforts were ultimately successful:

Most already knew how to act, and therefore, the company had to add fewer controls. The real effort involved making people aware of what compliance meant beyond each single rule (SEE1, a member of the supervisory body).

The CP practice prompted people to autonomously ensure that any future actions did not hide any attempt of bribery; this conveyed a *practical understanding*, which consisted of critically (self-)questioning each action's fairness from a legal and ethical point of view. Specifically, personnel could autonomously ensure that promotions, informative actions and various business projects avoided bribery towards doctors or other customers before these actions were executed:

Now when I have to make a decision... even basic, in daily activity (I mean CP is a matter of everything anybody does while working) [...] I think about the rules to see if that action is allowed, [and] if so, I ask myself if it is proper according to the legal and ethical point of view. Basically, I can say that before acting I think about what I am doing, but mostly I know that I am on the right way, because it is now almost natural (BE4, a pharmaceutical representative).

I used to pay attention to understanding what I had to do to stick to the procedures and to be unaffected after the controls. Now I am aware I am a part of the whole system of corporate responsibility, and I realise that I have to go beyond the single rule to behave in a proper way (BE3, a pharmaceutical representative).

We're linked to a precise code that allows us to stand out. Doctors know that we adhere to a strict work ethic, and therefore consider us a reliable resource. It is not only an honour for me to be working in a company that is seen as a benchmark in the industry, but it also gives credibility to my position and facilitates my work (BE4, a pharmaceutical representative).

This practical understanding is facilitated by introducing an additional *artefact* – an IT platform that plainly displayed the rules, explanations, examples of activities and compliance performance indicators. Such a platform was established to facilitate employees' awareness of the logic behind each rule and control. Moreover, managers sometimes review the procedures available in the IT platform to ensure that the CP's requirements will be fulfilled as they carry out their activities:

We have put in a lot of effort to help people understand that we have created an intranet platform where people can retrieve key points containing updates, executive summaries about the controls and a series of key performance indicators about compliance. Whenever somebody has a doubt, looking at the platform will provide advice, suggestions and examples. Of course, the compliance team is always available to help, but this platform helped people to understand (SE1, the compliance officer's assistant).

Supported by the board of directors and CEO, the compliance department continued to deliver periodic training sections and enriched them with discussions of the two prior lawsuits involving Pharma 101. Although the company successfully defended itself by demonstrating its proper conduct, these events allowed Pharma 101 to clarify the usefulness of its rules and procedures and provide concrete examples of proper and improper behaviours, as well as how to distinguish them. These training sessions prompted participants to realise not only the severity of the eventual consequences of misconduct but also the CP's usefulness in preventing them. Business managers and employees began to perceive the CP as a benefit to increase their confidence about identifying the "proper" action towards compliance. The interviewees' comments revealed that they felt confident in identifying, choosing and implementing actions to safeguard themselves and the company relative to the liabilities arising from LD 231. For example, the marketing manager of the oncology business unit (BM3) recognised that she and her colleagues had changed their mindset, which reflects their understanding of the added value for themselves as well as the company:

[Now] I feel confident that respecting the CP I am acting for the best. This makes me feel safe, as a worker, as a woman, as a wife and as a mum (BM3, marketing manager of the oncology business unit).

After observing a concrete example about the risk of lawsuits after misconduct, Pharma 101's personnel began to recognise the CP was paramount in preventing regrettable consequences. Moreover, managers and employees also realised the CP's usefulness for the company as well as their own credibility, such that their jobs became safer. They understood the CP not only in terms of avoiding misconduct but also as a responsibility to do their utmost to behave legally. The *general understanding* of the CP practice is reflected in legal-minded behaviour in conjunction with the new *teleoaffective structure* instilled by the CEO, supervisory board and

compliance staff. The latter entities realised that all CP-related activities must focus on promoting fair behaviour, and the organised training initiatives particularly aimed at increasing people's attitudes towards fair behaviour. Consequently, business managers and employees actively participated in the training activities organised by the compliance staff in collaboration with business departments. The compliance officer's assistant (SE1) noted that no one would miss those meetings because a common curiosity and interest exists regarding the topics pertaining the CP: (1) before the meeting, by preparing questions about new projects to ask the compliance officer; (2) during the meeting, by taking notes and memos during the presentation and (3) after the meeting, by discovering links between what the compliance officer said and the employee's own activities.

In summary, the practice during the third CP implementation phase consisted of employees' autonomous self-assessment of their daily activities relative to their legality. Critically (self-)questioning each action's fairness displayed their *practical understanding* of the CP practice at this stage, while the *general understanding* evolved towards safeguarding each action's corporate legality. This attitude is combined with a *teleoaffective structure* of fair behaviour, reflected as the intended aim and commitment towards fairness as an anticipated behaviour. The IT platform and periodic training sections are *artefacts* that facilitate further changes in the employees' practical and general understanding of the CP. This is coherently combined with the new teleoaffective structure that allows the previously mentioned artefacts to diffuse among all business managers and employees.

6. Discussion

This section provides a discussion of the findings in the light of the theoretical framework and points out the major contributions to the literature.

6.1 The mutual constitution of bractical and general understandings

Prior CG research has elucidated the practice lens approach's potential,l and Schatzki's account in particular, in capturing the actual functions of CG (Ahrens *et al.*, 2010, 2011; Ahrens and Khalifa, 2013; Brennan and Kirwan, 2015). Furthermore, Ahrens *et al.* (2010, 2011) argued that CG functioning strongly relies on practical understanding as "complexes of know-how [that] go beyond the individual's abilities to deal with aspects of corporate governance" (Ahrens *et al.*, 2011, pp. 319-320). The present research has drawn on this intuition to provide theoretically informed case insights.

The empirical analysis confirms the argument developed by Ahrens *et al.* (2010, 2011) by demonstrating that CP implementation strongly relies on practitioners' understanding of what to do to execute this CG mechanism. The research also offers an evolutionary perspective on the practical understanding by revealing how such understanding evolves over time. While the practical understanding was initially limited to executing what was explicitly indicated in the CP's rules, it has evolved to include executing the instructions gathered by company's CP experts – in this case, Pharma 101's compliance staff. This ultimately manifests as an autonomous self-assessment of the fairness of each action. Schatzki (2001, 2002) claimed that practical understandings alone rarely determine actions related to the practices that entail multiple actions, projects and emotions (Schatzki labels them as 'integrative practices'). Our findings confirm this theoretical argument for CP practices, which necessitate further practice-organising principles. However, the practical understanding emerges as a fundamental aspect of CP functioning as it was observed that practitioners without such an understanding of the actions to be executed will continuously check for instructions.

We interpret this finding given the fact that practitioners – and particularly those engaged in business activities – may not possess any particular expertise on CG issues and

AAAJ 33,4

904

need time to understand what a CG practice entails in terms of their responsibilities and related actions to be executed. When they ultimately realise the practice's requirements relative to their daily tasks, they will become confident and more prone to engage with the practice. The finding confirms the argument by Ahrens *et al.* (2011, p. 320), as follows:

[p]ractical understandings cannot [...] arise from principles or rules in isolation, nor can they be imposed by regulators. In any particular practice the knowledge of how to do something well only emerges over time through sustained work shared across people.

Indeed, practical understanding is a property of "particular practices" and not "a sort of general know-how lying behind most or all of human behaviour that exhibits sensitivity to context" (Welch and Warde, 2017, p. 187, emphasis in the original).

The research also indicates that practitioners' general understanding is particularly important in organising CP activities. When practitioners enact the CP with confusion about its overarching system, the CP's implementation was incredibly formal and manifested as merely a respect for the restrictions explicitly established by the CP's rules. When general understanding evolved towards a fear of penalties, practitioners continuously asked for the compliance staff's advice. Finally, when the attitude towards the CP advanced to become a willingness to safeguard corporate legality, such practitioner attitudes manifested as a self-assessment of their activities relative to their legality. Therefore, the paper expands prior CG research based on Schatzki's work by detailing the role of general understanding, a practice-organising principle that Ahrens et al. (2010 and 2011) neglected [7].

Our attempt to elucidate how general understanding influenced action also revealed reciprocal influences between practical and general understandings. On the one hand, the general understanding influences practitioners' comprehension of how to execute the CP in their daily actions. For instance, confusion in the first phase regarding the CP's overarching system compelled business staff to only do what the rules explicitly indicated. During the second phase, the fear of penalties resulted in executing only what was suggested by the compliance staff. On the other hand, the progressive understanding of how to implement the CP is important in supporting a "broader" understanding of the CP and its usefulness for both the company and its employees. For example, business employees in the third phase became autonomous in choosing the "proper" actions to execute; this helped them approach their CP responsibilities with constructive attitudes. This finding solicits a further conceptual and practical exploration of the reciprocal influences between practical and general understandings, which parallels the call to further engage with these concepts and related connections (Caldwell, 2012; Welch and Warde, 2017).

Prior research has interacted with the concept of general understanding in the context of accounting and risk management practices (Jørgensen and Messner, 2010; Nama and Lowe, 2014; Bui et al., 2019). Furthermore, Nama and Lowe (2014) have outlined the "mutually constitutive nature" of both general understanding and teleoaffective structures in the context of private equity practices. They particularly demonstrated a close connection between the attitude with which a practice is performed and the practices' affectivity. Bui et al. (2019) further explored this relationship in the context of risk management practices to reveal that general understanding is also strictly connected to the practice's teleological dimension. Our findings confirm these theoretical advancements by indicating that the general understanding coherently evolves with both the teleological and affective dimensions of a practice. Moreover, most practitioners' confusion during the first phase about the CP's overarching system reflects their assigned aim of (merely) adhering to the law and best practices. Furthermore, the fear of penalties during the second phase manifested as an intended aim and behaviour towards avoiding misconduct and a concern with incorrect actions. Finally, their attitude during the third phase revealed a willingness to safeguard

corporate legality, which matches the assigned aim and anticipated attitude of fair behaviour and commitment towards fairness.

6.2 The practitioners' immersion in the practice

Previous studies argued that "without a top management commitment to ethics, a company's ethics practices may be the kind that employees can easily ignore [...] executive commitment is essential" (Weaver et al., 1999a, b, p. 550). The current analysis confirms this argument, and our findings indicate that weak engagement by governance bodies and compliance staff with the CP's affective structure undermines the possibility to align the business staff's mindset with the practice's intended aims. Further, the CEO's sensitivity to compliance and business ethics favours the CP's enhancement. Employees' actual engagement with the CP requires not only top management's commitment but also a means to align practical and general understandings with the CP's requirements to overcome the barriers to behavioural changes that CP often face when inappropriate incentives or penalties are encountered (Weaver, 2014).

Therefore, the findings extend prior CP studies focussed on individuals' motivations, perceptions and incentives (Kaptein, 2011a, b; Singh, 2011) by demonstrating how practitioners' attitudes vary as result of their immersion in daily practice-related activities. Thus, the "effectiveness" (Kaptein, 2011a; Singh, 2011) of corporate codes of conduct cannot be ascribed to single variables but should be interpreted as an outcome of the interactions among practitioners. Furthermore, the findings demonstrate that changes in the rules affect the multiple activities and controls employees perform at different levels. However, how employees actually executed the CP depended more strongly on their overall and practical understanding of CP. The introduction of new rules represented only a first step towards affecting employees' mindset, as clearly highlighted in the insights from the case study's first phase. Without a comprehensive understanding of the responsibilities associated with the CP, it would have been difficult to widely implement this mechanism.

6.3 The role of artefacts in moulding bractices

Schatzki (2001, 2005) asserts that people mobilise artefacts in their efforts to implement practices. Furthermore, Ahrens *et al.* (2010, p. 6) posit that CG practices "stretch over diverse arrangements of people, artefacts, organisms and things". Artefacts in our research were fundamental in fostering practitioners' "immersion" in the CP practice. Various types of artifacts – such as internal control systems, training courses, performance indicators and formal and informal means of communication – emerged as particularly useful for encouraging interactions among different members involved in executing the CP. These artefacts fostered a widespread understanding of what comprised the CP in terms of daily activities and the responsibilities arising from its adoption. In our case study, adopting an IT platform in the third phase forged a practical understanding by facilitating practitioners' line of reasoning in their daily execution of the CP. By providing plain explanations of rules and examples of activities, this artefact acted as a reference point to ensure that practitioners' actions complied with the CP's requirements.

Artefacts also facilitated changes in general understanding; specifically, practitioners' fear of withhold on salary, and other disciplinary penalties were enhanced by corrections towards compliance behaviour within the remuneration system as well as the internal disciplinary system. Furthermore, practitioners changed their attitudes towards the CP after training activities focused on explaining the relevance of "proper" CP in the case of lawsuits; consequently, they understood the CP's usefulness for the company as well as themselves. In summary, the analysis demonstrates how CP-related artefacts moulded practical and general understandings and favour the latter to align with teleoaffective structures. This enriches prior studies that have marginally engaged with artefacts (Ahrens *et al.*, 2011). We believe

AAAJ 33,4

906

that the exploration of how CG artefacts are conceived of and used by practitioners at different organizational levels may offer further insights on CG daily functioning. While prior studies (Kaptein, 2011a, b; Schwartz, 2004) have highlighted that training activities influence practitioners' perceptions of the CP, the present research reveals that this perception has evolved as employees have engaged with related artefacts.

7. Conclusion

This article investigated how firm practitioners implement CPs, or specifically, how they conceive of and execute their CP responsibilities. Our case study analysis involved a practice lens approach as well as an interpretation of interviews and documentary materials collected from an Italian company with prolonged compliance behaviours. Our interpretation of CP was also influenced by Schatzki's practice organisation framework, which denotes CP as a practice organised by rules, practical and general understandings and teleoaffective structures. The analysis highlights the evolution in the CP practice. A practical understanding of daily actions required to accomplish the CP and a general understanding of the responsibilities connected with the CP, such as the attitudes with which the CP is performed, are mutually constitutive and jointly favour this evolution. Both of these types of understanding are facilitated by the use of specific artefacts. These artefacts also allow aligning practitioners' general understanding with the CP's assigned objectives and the desired reactions to them (teleoaffective structure).

The article offers several contributions. First, this research confirms the appeal of using practice theory – Schatzki's account in particular – to explore governance issues; it enriches our understanding of CG practices by providing theoretically informed case study insights on implementing this CG mechanism. Indeed, to the best of our knowledge, this is the first study to experiment with Schatzki's practice lens approach within an empirical setting involving CG practices. While prior studies have underlined practical understanding's significance in moulding CG activities, the present study empirically confirms the argument. Our work also extends this argument by highlighting the reciprocal influences between practical and general understandings, as well as how artefacts mould these understandings and spread specific teleoaffective structures among practitioners. We elucidate the close connections between practical and general understandings to theoretically advance Schatzki's practice organisation framework. This also supplements previous studies that have illuminated the "opaque" concept of general understanding (Caldwell, 2012), particularly in the context of accounting and risk management practices (Nama and Lowe, 2014; Bui *et al.*, 2019).

Future research may attempt to speculate as to how practitioners differently organise various firm practices – such as those related to accounting, risk management or CG – within the same organisational context. Generally, this research offers a foundation to further experiment with Schatzki's account, both within and across accounting, risk management and CG practices. Furthermore, the analysis offers a complimentary perspective on CP, an interdisciplinary topic that is increasingly investigated according to various approaches, whether objectivist, interpretative or social-constructionist (Parker and Nielsen, 2011). Therefore, we contribute the to the interpretivist task of "describing and understanding a whole range or organisational perceptions of and behavioural responses to regulation", rather than "mapping "compliance" and "noncompliance" in terms of levels of compliance (as objects of objectivist research)" (Parker and Nielson, 2011, p. 7). This study also establishes some practical implications in line with practitioners' calls for more attention to the accountability side of CP (Volkov, 2015; Jones, 2017). By highlighting the empirical insights about practitioners' understandings of the responsibilities the CP establishes, this work calls on governance bodies and staff departments to substantially consider both practical and general

understandings while forging the CP's regime of accountability. These understandings are fundamental to prompt constructive engagement with the CP at different organisational levels. Rules and disciplinary penalties are not enough in order to pursue the practitioners' engagement with the CP implementation. Whilst constituting the ground of CP enactment, rules and disciplinary penalties alone may even hinder constructive engagement with this CG mechanism.

This study also has limitations. First, its conclusions could be biased by its unique, specific research context. As the case company is a subsidiary of a global healthcare company, compliance was highly relevant even before the adoption of the CP associated with LD 231. The empirical analysis revealed the impact of this CP's adoption on employees' perceptions and actions; however, it is not possible to conclusively state whether these perceptions and actions would have emerged in the absence of the pre-existing compliance culture. Its heavily regulated industry also encouraged Pharma 101 to adopt the CP, as this industry defined organisational members' behaviour. Thus, it is difficult to predict how the company would have behaved in the absence of specific industry regulations. Further research projects might investigate companies with no previous experience with corporate codes of conduct.

The interview process occurred at a time when organisational members had experienced several years of CP adoption, such that their perceptions of more remote times would be less clear than those of more recent events. Furthermore, more recent changes may have influenced perceptions of their preliminary experiences with the CP. The research design sought to minimise the effects of this potential bias, but additional research might pursue ways to gather organisational members' perceptions at the moment they arise. However, while this instantaneous data collection may avoid memory biases, it may result in working on raw and preliminary perceptions. Researchers could also address the role of practitioners who operate beyond organisational boundaries, such as regulatory agencies or shareholder associations (Ahrens *et al.*, 2011). Generally, the approach inspired by the 'practice turn' in social sciences may not only help to address these limitations but also derive additional stimuli.

Notes

- 1. Corporate responsibility violations span the following categories: offences against public officials; information technology crimes and unlawful data processing; crimes committed by criminal organisations; breaches of public trust; crimes of obstruction of industry and trading activities; corporate crimes; offences concerned with terrorism or the subversion of the democratic order; crimes of female genital mutilation; offences against individuals; market abuse crimes; manslaughter or serious or critical injuries committed in violation of applicable workplace health and safety laws and regulations; receiving, laundering and using money, assets or profits of illicit origin; crimes involving copyright infringement; incitement to not testify or to bear false testimony to judicial authorities; environmental crimes; the employment of a foreign citizen with an invalid residence permit and transnational crimes.
- We use the labels "staff department" and "business department" to distinguish organisational departments according to the staff and business areas of the company: the former area includes the departments that support the organisation with specialised advisory and support activities, while the latter area includes the departments that directly participate in the organisation's core work.
- This account of practice organisation is based on Schatzki's (2002, 2010) more recent works, while his
 prior accounts (Schatzki, 1996, 2001) lacked a dimension discussing "general understanding".
- This type of CP, aimed to deter the crimes specified in LD 231, is the "organisational, management and controls model".
- According to the Italian LD 231, courts faced with evidence of misconduct will charge the company with "criminal liability" by scrutinising whether the CP was "properly" implemented to

- determine whether (and how): (1) management adopted and efficiently put into practice, prior to the offence, a CP in line with all legal requirements; (2) the task of monitoring CP implementation was entrusted to a ("supervisory") body with independent decision-making and control powers and (3) any lack of organisational supervision over the individual who committed the offences has been reported.
- Following the first-round review reports, we developed a second round of interviews to enrich our understanding of the CP implementation process throughout the company's hierarchy.
- We interpret this omission in the light of their reference to Schatzki's (1996, 2001) practice organisation framework, which lacked the "general understanding" dimension; see also footnote 3.

References

- Ahrens, T. and Chapman, C.S. (2007), "Management accounting as practice", *Accounting, Organizations and Society*, Vol. 32 Nos 1-2, pp. 1-27.
- Ahrens, T. and Khalifa, R. (2013), "Researching the lived experience of corporate governance", Qualitative Research in Accounting and Management, Vol. 10 No. 1, pp. 4-30.
- Ahrens, T., Chapman, C.S. and Khalifa, R. (2010), "Practicing corporate governance", Sixth Asia Pacific Interdisciplinary Research in Accounting (APIRA) Conference, Sydney, 12 and 13 July 2010, available at: http://apira2010.econ.usyd.edu.au/conference_proceedings/APIRA-2010-225-Ahrens-Practising-corporate-governance.pdf.
- Ahrens, T., Filatotchev, I. and Thomsen, S. (2011), "The research Frontier in corporate governance", Journal of Management and Governance, Vol. 15 No. 3, pp. 311-325.
- Allegrini, M. and D'Onza, G. (2003), "Internal auditing and risk assessment in large Italian companies: an empirical survey", *International Journal of Auditing*, Vol. 7 No. 3, pp. 191-208.
- Arena, M. and Azzone, G. (2007), "Internal audit departments: adoption and characteristics in Italian companies", *International Journal of Auditing*, Vol. 11 No. 2, pp. 91-114.
- Arjoon, S. (2005), "Corporate governance: an ethical perspective", Journal of Business Ethics, Vol. 61 No. 4, pp. 343-352.
- Auerbach, C. and Silverstein, L. (2003), Qualitative Data: An Introduction to Coding and Analysis, New York University Press, New York, NY.
- Bartrum, T.E. and Bryant, L.E. Jr (1997), "The brave new world of health care compliance programs", Annals of Health Law, Vol. 6 No. 1, pp. 51-75.
- Beale, S.S. and Safwat, A.G. (2004), "What developments in Western Europe tell us about American critiques of corporate criminal liability", *Buffalo Criminal Law Review*, Vol. 8 No. 1, pp. 89-163.
- Brennan, N.M. and Kirwan, C.E. (2015), "Audit committees: practices, practitioners and praxis of governance", *Accounting, Auditing & Accountability Journal*, Vol. 28 No. 4, pp. 466-493.
- Brennan, N.M. and Solomon, J. (2008), "Corporate governance, accountability and mechanisms of accountability: an overview", *Accounting, Auditing & Accountability Journal*, Vol. 21 No. 7, pp. 885-906.
- Brown, C. (2012), "The 'practice turn', phronesis and classical realism: towards a phronetic international political theory?", *Millennium-Journal of International Studies*, Vol. 40 No. 3, pp. 439-456.
- Bui, B., Cordery, C.J. and Wang, Z. (2019), "Risk management in local authorities: an application of Schatzki's social site ontology", The British Accounting Review, Vol. 51 No. 3, pp. 299-315.
- Caldwell, R. (2012), "Reclaiming agency, recovering change? An exploration of the practice theory of theodore schatzki", *Journal for the Theory of Social Behaviour*, Vol. 42 No. 3, pp. 283-303.
- Cetina, K.K., Schatzki, T.R. and von Savigny, E. (Eds) (2001), *The Practice Turn in Contemporary Theory*, Routledge, London.

- Corbella, S. (2013), I Modelli 231: La Prospettiva Aziendale. Una Proposta Interpretativa Nel Quadro Della New Institutional Theory, FrancoAngeli, Milano.
- Dukes, M.N. (2002), "Accountability of the pharmaceutical industry", Medicines, The Lancet, Vol. 360, pp. 1682-1684.
- Feldman, M.S. and Orlikowski, W.J. (2011), "Theorizing practice and practicing theory", Organization Science, Vol. 22 No. 5, pp. 1240-1253.
- Francis, R. and Armstrong, A. (2003), "Ethics as a risk management strategy: the Australian experience", *Journal of Business Ethics*, Vol. 45 No. 4, pp. 375-385.
- Gendron, Y. (2018), "Beyond conventional boundaries: corporate governance as inspiration for critical accounting research", Critical Perspectives on Accounting, Vol. 55, pp. 1-11.
- Hambrick, D.C., Werder, A.V. and Zajac, E.J. (2008), "New directions in corporate governance research", Organization Science, Vol. 19 No. 3, pp. 381-385.
- Hui, A., Schatzki, T. and Showe, E. (Eds) (2017), The Nexus of Practices. Connections, Constellations, Practitioners, Routledge, London.
- Jones, D. (2017), "Accountability: the shifting landscape of compliance responsibilities", available at: http://polarismanagement.com/accountability-shifting-landscape-compliance-responsibilities/.
- Jørgensen, B. and Messner, M. (2010), "Accounting and strategising: a case study from new product development", Accounting, Organizations and Society, Vol. 35 No. 2, pp. 184-204.
- Kaptein, M. (2011a), "Toward effective codes: testing the relationship with unethical behavior", Journal of Business Ethics, Vol. 99 No. 2, pp. 233-251.
- Kaptein, M. (2011b), "Understanding unethical behavior by unraveling ethical culture", Human Relations, Vol. 64 No. 6, pp. 843-869.
- Lai, A., Leoni, G. and Stacchezzini, R. (2019), Accounting and governance in diverse settings an introduction, Accounting History, Vol. 24 No. 3, pp. 325-337.
- MacLean, T.L. and Behnam, M. (2010), "The dangers of decoupling: the relationship between compliance programs, legitimacy perceptions, and institutionalized misconduct", Academy of Management Journal, Vol. 53 No. 6, pp. 1499-1520.
- McNulty, T., Zattoni, A. and Douglas, T. (2013), "Developing corporate governance research through qualitative methods: a review of previous studies", *Corporate Governance: An International Review*, Vol. 21 No. 2, pp. 183-198.
- Miettinen, R., Samra-Fredericks, D. and Yanow, D. (2009), "Re-turn to practice: an introductory essay", Organization Studies, Vol. 30 No. 12, pp. 1309-1327.
- Nama, Y. and Lowe, A. (2014), "The 'situated functionality' of accounting in private equity practices: a social 'site' analysis", Management Accounting Research, Vol. 25 No. 4, pp. 284-303.
- OECD (Organisation for Economic Co-operation and Development) (2007), "Glossary of statistical terms", available at: ec.europa.eu/eurostat/ramon/coded_files/OECD_glossary _stat_terms.pdf.
- Orlikowski, W.J. (2002), "Knowing in practice: enacting a collective capability in distributed organizing", *Organization Science*, Vol. 13 No. 3, pp. 249-273.
- Parker, C. (2006), "The 'compliance' trap: the moral message in responsive regulatory enforcement", Law & Society Review, Vol. 40 No. 3, pp. 591-622.
- Parker, L. (2018), "Corporate governance", in Roslender, R. (Ed.), *The Routledge Companion to Critical Accounting*, Routledge, New York, NY.
- Parker, C. and Nielsen, V.L. (2011), "Introduction", in Parker, C. and Nielsen, V.L. (Eds), *Explaining Compliance: Business Responses to Regulation*, Edward Elgar Publishing, Cheltenham.
- PricewaterhouseCooper (PwC) (2017), D.Lgs. 31/01. Indagine Nell'ambito Delle Società Quotate, PricewaterhouseCooper, available at: https://www.pwc.com/it/it/services/risk-assurance/assets/docs/pwc_dlgs_231_2016.pdf.

- Reckwitz, A. (2002), "Toward a theory of social practices: a development in culturalist theorizing", European Journal of Social Theory, Vol. 5 No. 2, pp. 243-263.
- Roberts, J., McNulty, T. and Stiles, P. (2005), "Beyond agency conceptions of the work of the non-executive director: creating accountability in the boardroom", British Journal of Management, Vol. 16, pp. S5-S26.
- Rossignoli, F. (2013), "Italian compliance programmes in groups of companies", *International Journal of Auditing Technology*, Vol. 1 Nos 3-4, pp. 294-311.
- Schatzki, T.R. (1996), "Practices and actions. A wittgensteinian critique of Bourdieu and Giddens", History of Human Sciences, Vol. 13 No. 3, pp. 283-308.
- Schatzki, T.R. (2001), "Practice mind-ed orders", in Cetina, K.K., Schatzki, T.R. and von Savigny, E. (Eds), The Practice Turn in Contemporary Theory, Routledge, London, pp. 50-63.
- Schatzki, T.R. (2002), The Site of the Social: A Philosophical Exploration of the Constitution of Social Life and Change, Pennsylvania State University Press, University Park, PA.
- Schatzki, T.R. (2005), "Peripheral vision. The sites of organizations", Organization Studies, Vol. 26 No. 3, pp. 465-484.
- Schatzki, T.R. (2006), "Peripheral vision. On organizations as they happen", Organization Studies, Vol. 27 No. 12, pp. 1863-1873.
- Schatzki, T.R. (2010), The Timespace of Human Activity: On Performance, Society, and History as Indeterminate Teleological Events, Lexington Books, New York, NY.
- Schatzki, T.R. (2018), "On practice theory, or what's practices got to do (got to do) with it?", in Edwards-Groves, C., Grootenboer, P. and Wilkinson, J. (Eds.), Education in an Era of Schooling. Critical Perspectives of Educational Practice and Action Research. A Festschrift for Stephen Kemmis, Springer, New York, NY, pp. 151-165.
- Schwartz, M.S. (2004), "Effective corporate codes of ethics: perceptions of code users", *Journal of Business Ethics*, Vol. 55 No. 4, pp. 321-341.
- Shah, N. and Napier, C. (2019), "Governors and directors: competing models of corporate governance", Accounting History, Vol. 24 No. 3, pp. 383-401.
- Shove, E. and Pantzar, M. (2005), "Consumers, producers and practices: understanding the invention and reinvention of nordic walking", *Journal of Consumer Culture*, Vol. 5 No. 1, pp. 43-64.
- Shove, E. and Southerton, D. (2000), "Defrosting the freezer: from novelty to convenience: a narrative of normalization", *Journal of Material Culture*, Vol. 5 No. 3, pp. 301-319.
- Simpson, B. (2009), "Pragmatism, mead and the practice turn", Organization Studies, Vol. 30 No. 12, pp. 1329-1347.
- Singh, J.B. (2011), "Determinants of the effectiveness of corporate codes of ethics: an empirical study", Journal of Business Ethics, Vol. 101 No. 3, pp. 385-395.
- Strauss, A. and Corbin, J. (1990), Basics of Qualitative Research, Sage, Newbury Park, CA, Vol. 15.
- Svensson, G., Wood, G., Singh, J. and Callaghan, M. (2009), "Implementation, communication and benefits of corporate codes of ethics: an international and longitudinal approach for Australia, Canada and Sweden", *Business Ethics: A European Review*, Vol. 18 No. 4, pp. 389-407.
- Volkow, M. (2015), "Compliance program effectiveness requires accountability", available at: http://blog.volkovlaw.com/2015/03/compliance-program-effectiveness-requires-accountability.
- Weaver, R.K. (2014), "Compliance regimes and barriers to behavioral change", Governance, Vol. 27 No. 2, pp. 243-265.
- Weaver, G.R. and Treviño, L.K. (1999), "Compliance and values oriented ethics programs: influences on employees' attitudes and behavior", Business Ethics Quarterly, Vol. 9 No. 2, pp. 315-335.

Weaver, G.R., Treviño, L.K. and Cochran, P.L. (1999), "Corporate ethics programs as control systems: influences of executive commitment and environmental factors", Academy of Management Journal, Vol. 42 No. 1, pp. 41-57.

Compliance program practices

Weaver, G.R., Treviño, L.K. and Cochran, P.L. (1999), "Integrated and decoupled corporate social performance: management commitments, external pressures, and corporate ethics practices", *Academy of Management Journal*, Vol. 42 No. 5, pp. 539-552.

911

- Welch, D. and Warde, A. (2017), "How should we understand 'general undertstandings'?", in Hui, A., Schatzki, T. and Showe, E. (Eds), *The Nexus of Practices. Connections, Constellations, Practitioners*, Routledge, London, pp. 183-196.
- Whittington, R. (2006), "Completing the practice turn in strategy research", Organization Studies, Vol. 27 No. 5, pp. 613-634.
- Whittington, R. (2011), "The practice turn in organization research: towards a disciplined transdisciplinarity", *Accounting, Organizations and Society*, Vol. 36 No. 3, pp. 183-186.

Corresponding author

Riccardo Stacchezzini can be contacted at: riccardo.stacchezzini@univr.it